Author’s response to reviews

Title: Undifferentiated epithelioid sarcoma presenting as a fever of unknown origin: A case report

Authors:

Nicholas Sajko (nick.sajko@dal.ca)
Allen Tran (Allen.Tran@nshealth.ca)
Shannon Murphy (sh437509@dal.ca)

Version: 1 Date: 30 Nov 2018

Author’s response to reviews:

Dear Reviewers,

As per the decision letter regarding this case report, changes have been made to address the minor edits brought forth by Reviewer #3.

As per Reviewer #3's request:

1. Ethnicity - Patient ethnicity was added to the case description outlined in page 5 of the case report, as well as in the abstract. The patient ethnicity can be identified in the revised case report by yellow highlighting.

2. Reference for page 3, line 14-17: "This diagnosis is not only rare in terms of sarcoma incidence rates, but also because of its unusual systemic presentation." - A reference for this line has now been added. Please find the sentence/claim in question highlighted in yellow on page 3, line 14-17.

Thank you.